Ohio Northern University DigitalCommons@ONU

Pharmacy Faculty Scholarship

Pharmacy

9-6-2017

Relationship between self-efficacy and patient knowledge on adherence to oral contraceptives using the Morisky Medication Adherence Scale (MMAS-8)

Daniel M. Tomaszewski *Chapman University*

Benjamin D. Aronson Ohio Northern University, b-aronson.1@onu.edu

Margarette L. Kading University of Minnesota - Duluth

Donald E. Morisky University of California, Los Angeles

Follow this and additional works at: https://digitalcommons.onu.edu/phar_faculty Part of the Epidemiology Commons, Other Pharmacy and Pharmaceutical Sciences Commons, and the Reproductive and Urinary Physiology Commons

Recommended Citation

Tomaszewski, D., Aronson, B. D., Kading, M., & Morisky, D. (2017). Relationship between self-efficacy and patient knowledge on adherence to oral contraceptives using the Morisky Medication Adherence Scale (MMAS-8). Reproductive health, 14(1), 110. doi:10.1186/s12978-017-0374-6

This Article is brought to you for free and open access by the Pharmacy at DigitalCommons@ONU. It has been accepted for inclusion in Pharmacy Faculty Scholarship by an authorized administrator of DigitalCommons@ONU. For more information, please contact digitalcommons@onu.edu.

RESEARCH

Open Access



Relationship between self-efficacy and patient knowledge on adherence to oral contraceptives using the Morisky Medication Adherence Scale (MMAS-8)

Daniel Tomaszewski¹, Benjamin D. Aronson^{2*}, Margarette Kading³ and Donald Morisky⁴

Abstract

Background: Preconception care, including family planning, is a vital component of healthcare for women of reproductive age. An average female spends the majority of her reproductive life trying to prevent a pregnancy. In order to prevent unintended pregnancy, women often rely on the use of hormonal contraceptives. In the United States, the majority of hormonal contraceptive users are prescribed oral contraceptive pills (OCPs). Reduced adherence to OCPs decreases their ability to prevent pregnancy. The study aimed to measure OCP adherence among female college students, and explore the relationship between OCP adherence, knowledge, and self-efficacy.

Methods: This cross-sectional study recruited a random sample of female college students to participate in an online survey. OCP adherence was based on the 8-item Morisky Medication Adherence Scale (MMAS-8). Secondary reporting of medication adherence included participant reports of the number of missed OCP doses in the previous month and typical month of use.

Results: Of the 5000 invited, 1559 (31.3%) completed the survey. Of those responding, 670 (41.3%) reported use of OCPs. A total of 293 (44.3%) OCP users met criteria for low adherence, 241 (36.4%) met criteria for medium adherence, and 128 (19.3%) met criteria for high adherence. Those with high adherence had higher self-efficacy (P < 0.001) and perceived knowledge (p < 0.001). After controlling for other factors, self-efficacy (b = .37) and perceived knowledge (b = .09) remained associated with OCP adherence.

Conclusion: Less than 20% of respondents met the criteria for high adherence to OCPs. Self-efficacy and knowledge were associated with higher OCP adherence. Targeted interventions from healthcare providers, health educators, and other adherence related media to increase the knowledge and self-efficacy of patients using OCPs may improve adherence rates. Additional research is needed to evaluate the impact of innovative interventions focused on social and behavioral patient factors, like knowledge and self-efficacy, on adherence to OCPs.

Keywords: Oral contraceptive adherence, Self-efficacy, Knowledge, Oral contraceptive pill, College students

* Correspondence: b-aronson.1@onu.edu

Full list of author information is available at the end of the article



© The Author(s). 2017 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

 $^{^2 \}rm Ohio$ Northern University Raabe College of Pharmacy, 525 S Main St, Ada, OH 45810, USA

Plain English summary

A large portion of women rely upon oral contraceptive pills (OCPs) as a part of their family planning strategy. For OCPs to work effectively at preventing pregnancies, women must consistently take them. In other words, OCPs require good adherence. In this study women attending college at one university were recruited to answer questions in an online survey to detect any problems with OCP adherence, and determine if self-efficacy or knowledge influence OCP adherence.

The women that participated in the online survey were asked if they used OCPs. Those who used OCPs were asked about their self-efficacy, or feeling that they could confidently take their birth control correctly, knowledge of their OCP, and adherence in taking their OCP. In addition, the women in this study provided their age, relationship status, reason for OCP use, and past use of emergency contraception.

A total of 670 women, or just over 40% of those in this study, used OCPs. Over 50% of these women missed at least one dose of their OCP in the month prior to the survey. Adherence to OCPs was related to both selfefficacy and knowledge.

In sum, there is room to improve OCP adherence in women attending college. Increasing knowledge about OCPs and building self-efficacy may be effective strategies to improve OCP adherence.

Background

Preconception care, including family planning, is a vital component of healthcare for women of reproductive age. Unsuccessful pregnancy prevention efforts have the potential to lead to unintended pregnancies. The consequences of an unintended pregnancy, which include those that were mistimed and unwanted, include impacts to the fetus, the mother, and society. Fetal impacts include a delay in receiving necessary prenatal supplements, potential for exposure to environmental risks (including alcohol, nicotine, or other drugs), and increased potential for birth defects and low birth weight [1]. Additionally, children resulting from unintended pregnancies have been shown to experience poor mental and physical health and have lower educational attainment and more behavioral health issues in their teens [2]. Mothers of unintended pregnancies have been shown to be at increased risks of adverse effects from delays in prenatal care, maternal depression, and reduction in income potential [3]. Lastly, a significant burden has been placed on society by unintended pregnancies. Estimates suggest that unintended pregnancies cost the United States \$21 billion in 2010, which includes the cost of births, abortions, and miscarriages from unintended pregnancies [4].

Although only 8% of U.S. women report unmet family planning needs and 99.1% of sexually active women report having used some form of contraceptive, there remains significant disparities in pregnancy prevention [5, 6]. Prior to 2011 there had been nearly no change over the last 30 years in pregnancies that were reported as unintended when conceived (49%) or births that were reported to be unintended (37%) [7–9]. This is particularly troubling, as the availability of contraceptives has improved over this same timeframe [9]. Although these rates dropped in 2011 to 45% and 36%, respectively, there remains a considerable concern over the high unintended rate and the prevention of unintended pregnancy continues to be a key focus of the public health system in the United States [1].

Consistent and correct use of modern contraception has a dramatic impact on unintended pregnancies. Only 68% of women that are at risk for an unintended pregnancy report correct and consistent use of contraception, and these women account for only 5% of unintended pregnancies [10]. Estimates suggest that non-adherence may be directly the cause of approximately 20% of unintended pregnancies each year [11]. For oral hormonal contraceptives (also known as "the pill" or oral contraceptive pill, OCP), perfect use requires users to take every dose within the same 4-h window each day. Those who are not able to maintain perfect use, or are non-adherent to their contraception method of choice, reduce the effectiveness rates considerably [11]. Perfect adherence has been shown to prevent nearly all unintended pregnancies (<1% failure rate), compared to a nearly 9% failure rate with typical use (i.e., inconsistent or incorrect use) [12, 13].

In this study, less than perfect adherence is referred to as poor adherence, or nonadherence. Past estimates of OCP nonadherence range from 25% not taken within the "window of hormonal safety" [14], to approximately 33–52% of women missing 1 or more pill per cycle [12, 15–17], with over 20% missing 2 or more pills per cycle [16]. Even in an oral contraceptive clinical trial, the rate of nonadherence (labelled noncompliance in the study) was 17% [18]. It is important to consider potential causes of nonadherence and interventions that may appropriately target these causes.

Previous research has suggested a number of factors associated with poor OCP adherence, but the research is limited overall [19–21]. Previously identified correlates of OCP adherence include the establishment of a pill taking routine, information seeking habits, and race and ethnicity [19, 20]. In addition, one study showed higher levels of patient reported perceived behavioral control and the presence of coping strategies for non-adherence were positively associated with adherence to OCPs [21].

Given the limited data regarding influential factors of OCP adherence, this study aims to investigate several possible determinants. The primary objectives of this study are to: 1) determine self-reported adherence to OCPs based upon a validated measure, 2) explore demographic differences in adherence rates, 3) examine the relationship between self-efficacy and perceived knowledge on adherence.

Methods

This cross-sectional research was conducted using an online survey platform, Qualtrics[®]. The survey was electronically distributed to participants using their preferred email. The online survey platform allowed complex skip patterns and flow logic to be employed, whereby a respondent's earlier choice in the survey determined the question displayed, and also the specific OCP product referenced in the question. Participants not currently using any form of hormonal contraceptive were directed to answer a separate set of questions designed to gather basic demographics, involvement in risky behaviors, and method of pregnancy prevention used. The University of Minnesota and Chapman University Institutional Review Boards approved this study.

Participants

This study was conducted at the University of Minnesota, a public research university in the Midwestern United States with approximately 50,000 thousand enrolled students, of which over 25,000 are female. A simple random sample of 5000 female students was obtained from the University of Minnesota's Office of Institutional Research. This sample was limited to the list of degree-seeking females with full-time enrollment status aged 18 and older. Students who had suppressed their contact information were excluded from the panel. A total of 5000 email invitations were sent to potential participants, 14 of which were returned as undeliverable. One reminder e-mail was sent 7 days after the initial e-mail to selected individuals that had not yet responded. In sum, 1750 individuals opened the survey, 1623 consented to participate, and 1559 completed the survey. The survey completion rate of invited individuals who received the invitation email was 31.3% (1559/4986). Only those taking OCPs were included in the analyses below (n = 670).

Measures

OCP Adherence

Adherence to OCPs was measured using the 8-item Morisky Medication Adherence Scale (MMAS-8) [22]. This scale is extensively used in survey research as a proxy for medication adherence, and has shown concurrent validity with non-humanistic markers of adherence [22-25]. The items in the scale ask participants guestions about their medication-related behaviors and causal indicators, for example, "Do you sometimes forget to take your oral contraceptive?" The first seven items have a yes/no response format. The final item, "How often do you have difficulty remembering to take your oral contraceptive?" A five-item Likert scale was used to measure item 8.. Items were summed to provide a continuous measure of selfreported adherence ranging from 0 to 8, with higher scores indicating higher adherence. No participants had missing data on more than one MMAS-8 item, thus all met the 75% eligibility criterion for the scale. Per standard scoring procedures, missing data on MMAS-8 items were imputed by using the median value from those participants who met the eligibility criteria. Cut points were used to categorize high (sum score = 8), medium (sum score 6 to <8), and low (sum score < 6) adherence based on previously established categories [22]. In the original study of patients with high blood pressure, high adherence was associated with blood pressure control, which further demonstrated the criterion-related validity of the MMAS-8 in measuring adherence [22]. In order to compare the present findings to prior studies, we also asked how many days participants forgot to take their OCP in a typical month, and then more specifically last month. Response options were Never, 1 time per month, 2–3 times per month, 4–5 times per month, and 6 or more times per month. Self-report of missed doses was used to describe the sample, and compared by adherence category.

Correlates of OCP Adherence

Perceived knowledge of OCP being used was measured using one item. Perceived knowledge is defined here as a self-assessment of knowledge, or in other words, what each respondent thinks they know about their OCP. Participants were asked: How would you rate your knowledge of your hormonal contraceptive, on a scale from 1 to 10, where 1 means "I know nothing" and 10 means "I know everything." Responses were recoded to be grounded at zero (i.e., ranging from 0 to 9), with a higher score equating to higher perceived knowledge. Self-efficacy of using OCP correctly was measured with one item. Here, self-efficacy is defined as the belief that one can successfully perform a behavior [26], namely in this study, correctly use her OCP. We measured self-efficacy by asking individuals to indicate their level of agreement on a 5-point Likert-type scale with the statement "I feel confident that I use my contraceptive correctly." The responses were coded from 0 ('Strongly Disagree') to 4 ('Strongly Agree').

Covariates

Participants reported their age, relationship status, reason for OCP use, and past emergency contraceptive use. Age was collected as a continuous variable. Participants indicated their current relationship status, and we dichotomized responses as in a committed relationship (married, committed dating relationship, engaged) or not in a committed relationship (single, separated, widowed, divorced). Participants were asked to report the reasons why they were using OCP (i.e., pregnancy prevention, help regulate menses, reduce symptoms associated with menses, improving acne, treatment of premenstrual dysphoric disorder, or other), which was used to create a new variable dichotomized by those indicating that pregnancy prevention was one of the reasons they were using OCP (1), and those that did not (0).

Statistical analysis

We first examined the descriptive statistics for all study variables. To examine potential differences in participant characteristics and bivariate relationships between study variables, we used ANOVA for continuous variables and chi square test for categorical variables. The alpha value for statistical significance was set at 0.05, with a Bonferroni correction applied for comparisons of continuous variables between adherence categories. We used ordinary least squares regression to determine the relationship between perceived knowledge, self-efficacy, and medication adherence after controlling for age, relationship status, and use of OCP for pregnancy prevention. List-wise deletion of cases with missing data resulted in 650 total observations for regression analysis.

Results

The overall sample contained 670 OCP users (41.9%), 228 other contraceptive users (14.3%), and 700 nonusers (43.8%). Of the OCP users, 293 (44.3%) met criteria for low adherence, 241 (36.4%) met criteria for medium adherence, and 128 (19.3%) met criteria for high adherence (Table 1). The mean score for OCP users' reported knowledge of their OCP was 6.03 (scored from 0 to 9), signifying a potential knowledge gap. Conversely, a majority of participants reported higher rates of self-efficacy, with a mean score of 3.43 (scored 0 to 4). Additional participant data and demographic characteristics are shown in Table 1.

Table 2 provides a comparison of demographics and correlates of adherence by OCP adherence category. Significant differences were found between groups for relationship status (P = 0.016), with the proportion in a relationship highest among the high adherence group. There were significant differences in perceived knowledge

Table 1 Demographic characteristics of participants using OCPs

	Mean/% (S.D.)
Age	21.95 (4.1)
Years in college	3.42 (2.1)
Relationship status	
In a relationship	61.2%
Not in a relationship	38.8%
Primary reason(s) for using hormonal contraception ^a	
Pregnancy prevention	84.6%
Regulation of menses	71%
Reduce symptoms associated with menses	52.7%
Improving acne	35.4%
Treatment of Premenstrual Dysphoric Disorder	2.5%
Other	3.7%
Sexual orientation	
Heterosexual	92.5%
Gay/Lesbian	0.8%
Bisexual	4.7%
Unsure	2%
Typical month missed doses	
Never	37.8%
1 time per month	39.6%
2–3 times per month	18.6%
4–5 times per month	2.7%
6 or more times per month	1.2%
Last month missed doses	
Never	46.9%
1 time per month	31.2%
2–3 times per month	17.9%
4–5 times per month	3.3%
6 or more times per month	0.8%
Mean perceived knowledge (range 0 to 9)	6.03 (1.5)
Mean perceived self-efficacy (range 0 to 4)	3.43 (0.6)

^aParticipants could select multiple responses; OCPs oral contraceptive pills; S.D. standard deviation

(P < 0.001) and perceived self-efficacy (P < 0.001). The low adherence group had significantly lower perceived knowledge compared to the medium (P = 0.002) and high (P < 0.001) adherence groups. Additionally, the low adherence group had significantly lower perceived selfefficacy compared to the medium and high groups (P < 0.001 for both), and the medium group had significantly lower self-efficacy than the high group (P = 0.028).

Results of ordinary least squares regression are shown in Table 3. After controlling for age, relationship status, and use of OCP for pregnancy prevention,

	Low adherence	Medium adherence	High adherence	Ρ
Overall (%)	44.3%	36.4%	19.3%	n/a
Age (mean) ^a	21.8	21.9	22.4	.346
Years in college (mean) ^a	3.4	3.3	3.6	.559
In a relationship (%) ^B	55.9%	61.7%	71.9%	.009
OCP for pregnancy prevention (%) ^b	84.2%	84.2%	87.5%	.652
Perceived knowledge (mean) ^a	5.74	6.18	6.40	<.001
Perceived self-efficacy (mean) ^a	3.21	3.54	3.71	<.001
2 or more missed pills typical month ^b	45.9%	5.8%	0.8%	<.001
2 or more missed pills last month ^b	42.7%	8.3%	0%	<.001

Table 2 Demographic characteristics and correlates of adherence by OCP adherence category

^aANOVA ^bchi-square test

Use of the ©MMAS is protected by US copyright laws. Permission for use is required. A Licensure agreement is available from: Donald E. Morisky, ScD, ScM, MSPH, Professor, Department of Community Health Sciences, UCLA School of Public Health, 650 Charles E. Young Drive South, Los Angeles, CA 90095–1772, dmorisky@ucla.edu

perceived knowledge (beta = .08; P = .032) and selfefficacy (beta = .37; P < .001) were both positively related to OCP adherence scores using the MMAS-8. The R² of the regression model was 0.16.

Discussion

This study determined the rate of OCP adherence, explored demographic differences in adherence rates, and examined the relationship between self-efficacy and perceived knowledge on adherence for a sample of women enrolled at a large Midwest university. According to the MMAS-8, 80.7% of OCP users exhibited some degree of non-adherence, or in other words less than perfect adherence, higher than past estimates of non-adherence to OCPs [12–18]. The report of missed doses in an average/typical month in this study was also higher than previously reported estimates (62% reported missing at least 1 dose per month, compared to 33–52% reported in previous research [12, 15–17]).

Several differences were noted between adherence groups when categorized by the MMAS-8. While there was no difference in age, years in college, or use of OCP for pregnancy prevention, a higher proportion of those in the high adherence group reported being in a relationship. This latter finding is consistent with other research demonstrating that women in consistent relationships are more likely to use effective contraception when compared to women in casual relationships [27]. As expected, proportionally fewer individuals in the high adherence group reported missing 2 or more doses in both a typical month and last month, and the high adherence group exhibited a higher mean perceived knowledge and higher mean perceived self-efficacy when compared to the low adherence group.

After controlling for demographic factors, perceived knowledge and self-efficacy remained positively related to OCP adherence. These findings, while cross-sectional, suggest that knowledge and self-efficacy could be useful targets for future interventional research focused on improving OCP adherence. This is supported by prior work that has shown the amount of accompanying educational material read and understood was related to adherence [19]. Additionally, a study by Molloy and colleagues established the relationship between perceived behavioral control, a concept similar to selfefficacy, and medication adherence among OCP users [21]. These findings in OCP adherence are consistent with a cadre of prior literature that established the relationship between self-efficacy on medication adherence for other disease states [28, 29].

Table 3 Results of ordinary least squares regression of OCP adherence score

	Unstandardized coefficient (B)	Standard error	Standardized coefficient (beta)	Р
Constant	2.45	.46		.000
Age	.00	.01	02	.641
In a relationship	.19	.12	.06	.116
OCP for pregnancy prevention	06	.16	01	.725
Perceived knowledge	.09	.04	.08	.032
Perceived self-efficacy	.93	.10	.37	.000
Model adjusted R^2	.16			

Current recommendations for counseling adolescents about contraception include assessment of self-efficacy toward a contraceptive method [30]. This recommendation is also shared within the CDC's and U.S. Office of Population Affairs' family planning recommendations, which recommends that the provider evaluates potential barriers to continued appropriate use of the contraception being considered [31]. These recommendations encourage the provision of counseling services to improve patient knowledge, but also highlight the importance of evaluating social-behavioral factors, including patient confidence in ability to use consistently and correctly [31]. Our findings support these recommendations.

Limitations

Although this study provides insight on factors influencing OCP adherence, the findings must be placed in context of the limitations. First, the use of a crosssectional method to collect data does not allow causality to be inferred from the results. In addition, the methodology employed in this study could lead to volunteer bias, and potential recall bias. Use of the MMAS-8 has historically focused on measuring adherence to medications related to chronic or long-term infectious conditions, and though OCP use is similar to other chronic medications (i.e. medication must be taken daily), it is unclear if the MMAS is the most appropriate medication adherence scale to use. Previous research has shown the ability to tailor the MMAS-8 to unique disease states and dosage forms, such as, asthma inhalants, injectable insulins for diabetes, and oral osteoporosis medications dosed less frequently than daily [32]. In this study, selfreport of missed doses was related to the MMAS-8 adherence categories, providing evidence for its use in OCP research. Further research and modification to the original scale may be warranted to develop an optimal scale. This study's cohort had a higher education level compared to the general population in the U.S., limiting the generalizability of these findings to the overall U.S. population. It is concerning that in this highly educated sample such high levels of missed OCP doses were found. Lastly, the measures of perceived knowledge and self-efficacy used in this study were single-item indicators. While these indicators could be helpful clinical tools in terms of ease of administration, answers do not specifically indicate what knowledge is lacking or what elements of self-efficacy are low. This limits the ability to more specifically examine possible reasons for OCP nonadherence, and design interventions to remedy these gaps. However, low perceived knowledge or self-efficacy reported by a patient could prompt a clinician to delve into tailored counseling to ameliorate knowledge gaps and improve self-efficacy.

Conclusion

Based upon the adherence measure used, only 20% of the women in this study met criteria for a high level of OCP adherence. The findings from multivariable analysis affirms the relationship between self-efficacy and perceived knowledge on OCP adherence. We are not aware of any other study that has linked perceived knowledge of OCPs to OCP adherence. These findings highlight the importance of patient knowledge and self-efficacy and the need to target these factors to improve adherence. Healthcare practitioners, health educators, and other adherence related media may need to consider focusing their attention to efforts aimed at ensuring proper patient knowledge and a better sense of self-efficacy in managing OCP use. More research is necessary to determine the best measure of OCP adherence and effective approaches to improve selfefficacy and knowledge of patients to their OCP. Additionally, future research needs to measure the impact of these approaches on adherence to OCPs and reduction in unintended pregnancies.

Abbreviations

ANOVA: Analysis of Variance; MMAS-8: Morisky Medication Adherence Scale (8-item); OCP: Oral Contraceptive Pill

Acknowledgements

We would like to acknowledge University of Minnesota's Boynton Health Services and the University of Minnesota College of Pharmacy for funding this research project.

The MMAS (8-item) content, names and trademarks are protected by US copyright and trademark laws. Permission for use of the scale and its coding is required. A license agreement is available from: Donald E. Morisky, ScD, ScM, MSPH,14725 NE 20th St Bellevue, WA 98007, USA; dmorisky@gmail.com.

Funding

The study was co-funded by the University of Minnesota College of Pharmacy Office of Research and University of Minnesota's Boynton Health Services.

Availability of data and materials

The datasets generated and/or analyzed during the current study are not publicly available as the University of Minnesota IRB approved the public release of the data collected during the study only in aggregate form but are available from the corresponding author on reasonable request.

Authors' contributions

All authors participated in the development and completion of the final manuscript. DT, BA, and MK were all involved in the conceptual development of the study. DM provided insight into the use of the MMAS-8 in the study population and also participated in the manuscript development. DT led the efforts in writing the final manuscript and BA led the statistical analysis efforts. All authors reviewed and approved the final manuscript.

Ethics approval and consent to participate

This study was reviewed and approved by the University of Minnesota and Chapman University's Internal Review Board. All participants explicitly gave their consent to participate in the study. Consent was collected electronically.

Consent for publication

Not Applicable.

Competing interests

Morisky receives a royalty for use of the copyrighted and trademarked MMAS-8.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Author details

¹Chapman University School of Pharmacy, 9401 Jeronimo Rd, Irvine, California 92618, USA. ²Ohio Northern University Raabe College of Pharmacy, 525 S Main St, Ada, OH 45810, USA. ³College of Pharmacy, University of Minnesota, 232 Life Sciences, 1110 Kirby Dr, Duluth, MN 55812, USA. ⁴UCLA Fielding School of Public Health, 650 Charles E. Young Dr. South, 16-035 Center for Health Sciences, Los Angeles, CA 90095-1772, USA.

Received: 4 March 2017 Accepted: 10 August 2017 Published online: 06 September 2017

References

- 1. Cheng D, Schwarz E, Douglas E, Horon I. Unintended pregnancy and associated maternal preconception, prenatal and postpartum behaviors. Contraception. 2009;79(3):194–8.
- Herd P, Higgins J, Sicinski K, Merkurieva I. The implications of unintended pregnancies for mental health in later life. Am J Public Health. 2016;106(3): 421–9.
- Wendt A, Gibbs CM, Peters S, Hogue CJ. Impact of increasing interpregnancy interval on maternal and infant health. Paediatr Perinat Epidemiol. 2012;26(Suppl. 1):239–58.
- Sonfield A, Kost K. Public Costs from Unintended Pregnancies and the Role of Public Insurance Programs in Paying for Pregnancy-Related Care: National and State Estimates for 2010. New York: Guttmacher Institute; 2015. http:// www.guttmacher.org/pubs/public-costs-of-UP-2010.pdf. Accessed Accessed 1 Dec 2016
- Sonfield A, Hasstedt K, Kavanaugh ML, Anderson R. The Social and Economic Benefits of Women's Ability To Determine Whether and When to Have Children. New York: Guttmacher Institute; 2013. https://www. guttmacher.org/report/social-and-economic-benefits-womens-abilitydetermine-whether-and-when-have-children. Accessed 1 Dec 2016
- Finer LB, Zolna MR. Unintended pregnancy in the United States: incidence and disparities, 2006. Contraception. 2011;84(5):478–85.
- United Nations Population Division, Department of Economic and Social Affairs. 2015 Update for the MDG Database: Unmet Need for Family Planning. http://www.un.org/en/development/desa/population/theme/ family-planning/index.shtml. Accessed 1 Dec 2016.
- Guttmacher Institute. Unintended Pregnancies in the United States. 2015. http://www.guttmacher.org/pubs/FB-Unintended-Pregnancy-US.pdf. Accessed 1 Dec 2016.
- Mosher WD, Jones J, Abma JC. Intended and unintended births in the United States: 1982–2010. In: National health statistics reports; no 55. Hyattsville: National Center for Health Statistics; 2012.
- Daniels K, Mosher WD, Jones J. Contraceptive methods women have ever used: United States, 1982–2010. In: National health statistics reports; no 62. Hyattsville: National Center for Health Statistics; 2013. https://www.cdc.gov/ nchs/data/nhsr/nhsr062.pdf. Accessed 1 Dec 2016.
- 11. Rosenberg M, Waugh M. Causes and Consequences of Oral Contraceptive Noncompliance. Am J of Obstet Gynecol. 1999;180(2):5276–9.
- Kost K, Singh S, Vaughan B, Trussell J, Bankole A. Estimates of contraceptive failure from the 2002 National Survey of Family Growth. Contraception. 2008;77(1):10–21.
- Trussell J. Contraceptive failure in the United States. Contraception. 2011; 83(5):397–404.
- 14. Oakley D, Sereika S, Bogue EL. Oral contraceptive pill use after an initial visit to a family planning clinic. Fam Plan Perspect. 1991;23:150–4.
- Rosenberg MJ, Waugh MS, Meehan TE. Use and misuse of oral contraceptives: risk indicators for poor pill taking and discontinuation. Contraception. 1995;51:283–8.
- Rosenberg MJ, Waugh MS. Improving oral contraceptive use: factors associated with poor compliance. Denver: Proceedings of American College of Obstetricians and Gynecologists Annual Meeting; 1996.
- Potter L, Oakley D, de Leon-Wong D, Caneonar R. Measuring compliance among oral contraceptive users. Fam Plan Perspect. 1996;28:154–8.
- Westhoff CL, Torgal AT, Mayeda ER, Shimoni NA, Stanczyk FZ, Pike MC. Predictors of noncompliance in an oral contraceptive clinical trial. Contraception. 2012;85(5):465–9.

- Rosenberg MJ, Waugh MS, Burnhill MS. Compliance, counseling and satisfaction with oral contraceptives: a prospective evaluation. Fam Plan Perspect. 1998:89–104.
- 20. Steinkellner A, Chen W, Denison SE. Adherence to oral contraception in women on Category X medications. Am J Med. 2010;123(10):929–34.
- Molloy GJ, Graham H, McGuinness H. Adherence to the oral contraceptive pill: a cross-sectional survey of modifiable behavioural determinants. BMC Public Health. 2012;12:838.
- Morisky DE, Ang A, Krousel-Wood M, Ward HJ. Predictive validity of a medication adherence measure in an outpatient setting. J Clin Hypertens. 2008;10(5):348–54.
- Morisky DE, Green LW, Levine DM. Concurrent and predictive validity of a self-reported measure of medication adherence and long-term predictive validity of blood pressure control. Med Care. 1986;24:67–74.
- Krousel-Wood MA, Islam T, Webber LS, Re RS, Morisky DE, Muntner P. New Medication Adherence Scale Versus Pharmacy Fill Rates in Seniors With Hypertension. Am J Manag Care. 2009;15(1):59–66.
- Morisky DE, DiMatteo MR. Improving the measurement of self-reported medication nonadherence: Final response. J Clin Epidemiol. 2011;64:258–63.
- Bandura A. Self-efficacy: toward a unifying theory of behavioral change. Psychol Rev. 1977;84(2):191.
- Upadhyay UD, Raifman S, Raine-Bennett T. Effects of relationship context on contraceptive use among young women. Contraception. 2016;94(1):68–73.
- Walker RJ, Smalls BL, Hernandez-Tejada MA, Campbell JA, Egede LE. Effect of diabetes self-efficacy on glycemic control, medication adherence, selfcare behaviors, and quality of life in a predominantly low-income, minority population. Ethn Dis. 2014;24(3):349–55.
- Breaux-Shropshire TL, Brown KC, Pryor ER, Maples EH. Relationship of blood pressure self-monitoring, medication adherence, self-efficacy, stage of change, and blood pressure control among municipal workers with hypertension. Workplace Health Saf. 2012;60(7):303–11.
- Jaccard J, Levitz N. Counseling adolescents about contraception: towards the development of an evidence-based protocol for contraceptive counselors. J Adolesc Health. 2013;52(4):56–513.
- Gavin L, Pazol K. Update: Providing Quality Family Planning Services Recommendations from CDC and the U.S. Office of Population Affairs, 2015. MMWR Morb Mortal Wkly Rep. 2016;65:231–4. http://dx.doi.org/10.15585/ mmwr.mm6509a3. Accessed 1 Dec 2016
- Reynolds K, Viswanathan HN, Muntner P, Harrison TN, Cheetham TC, Hsu JW, Gold DT, Silverman S, Grauer A, Morisky DE, O'Malley CD. Validation of the Osteoporosis-Specific Morisky Medication Adherence Scale in long-term users of bisphosphonates. Qual Life Res. 2014;23(7):2109–20.

Submit your next manuscript to BioMed Central and we will help you at every step:

- We accept pre-submission inquiries
- Our selector tool helps you to find the most relevant journal
- We provide round the clock customer support
- Convenient online submission
- Thorough peer review
- Inclusion in PubMed and all major indexing services
- Maximum visibility for your research

Submit your manuscript at www.biomedcentral.com/submit

